Author’s response to reviews

Title: Prevalence and correlates of suicidal ideation in Korean firefighters: A nationwide study

Authors:
Heyeon Park (ju-aa@hanmail.net)
Johanna Kim (iambabyvox@hanmail.net)
Beom Jun Min (mbj0310@gmail.com)
Sohee Oh (oh.sohee@gmail.com)
Jeong-Hyun Kim (retrial3@hanmail.net)

Version: 3 Date: 02 Dec 2019

Author’s response to reviews:

Thank you for taking your precious time to review our manuscript “Prevalence and correlates of suicidal ideation in Korean firefighters: a nationwide study.” We have made every effort to accommodate the feedback and the following changes were made in our manuscript to reflect your input.

1. Please include the supplementary tables (1 and 2) that are referenced in the manuscript.

=> We uploaded the supplementary tables (1 & 2).

2. We notice significant text overlap between your manuscript and previously published materials. Please reformulate the large sections of overlapping text present in your Methods section. Please ensure that, where relevant, these sources are also referenced as appropriate. “The impact of emotional labor on the severity of PTSD symptoms in firefighters”

https://doi.org/10.1016/j.comppsych.2018.03.002

=> We rewrote the overlapping texts in our Introduction and Methods.

3. Please add a Declarations heading after the Abbreviations section.

= We added a Declarations heading after the Abbreviations.
4. Please proofread and ensure that when you upload your revised submission it is in the final form for publication.

=> We proofread and ensure our revised manuscript.

Thank you, again.

We look forward to your positive reply.

Yours sincerely,

Dr. Jeong-Hyun Kim.

Department of Psychiatry
Seoul National University Bundang Hospital
Seongnam, South Korea
Phone: +82-31-787-2025
Email: retrial3@hanmail.net